**Author's response to reviews**

**Title:** Gender distribution of adult patients on highly active antiretroviral therapy (HAART) in Southern Africa: a systematic review

**Authors:**

Adamson S Muula (muula@email.unc.edu)
Thabale J Ngulube (thabalejackngulube@gmail.com)
Seter Siziya (ssiziya@yahoo.com)
Cecilia M Makupe (cmakupe@chanco.unima.mw)
Eric Umar (ericumar@yahoo.com)
Hans W Prozesky (hwp@sun.ac.za)
Charles S Wiysonge (wiysongeC@unaids.org)
Ronald H Mataya (rmataya@llu.edu)

**Version:** 6 **Date:** 21 April 2007

**Author's response to reviews:** see over
21 April 2007

Authors’ comments

1. We were asked to remove full stop at end of title of manuscript. This has been done.
2. Country of affiliation for all authors: This has also been done
3. Abstract is submission system to be the same as abstract in manuscript: This has been checked to be the same
4. Remove conflict of interest in preference of competing interests. This has been done
5. Website references to be removed from text: This has been done.
6. Website references to be formatted as advised. We have since removed that reference in preference to hard copy that we have referenced now.
7. Put Table 2 in Landscape format: This has been done.
8. All authors names in references to be listed unless reference has > 30 authors. We were not able to adhere to this in reference 34 because the abstract itself did not report the names of the other authors. We have however reported all authors’ names as was on the abstract.

Thank you Editor

Adamson Muula